Author's response to reviews

Title: Development and formative evaluation of the e-Health Implementation Toolkit (e-HIT).

Authors:

Elizabeth Murray (elizabeth.murray@ucl.ac.uk)
Carl May (c.r.may@newcastle.ac.uk)
Frances Mair (f.mair@clinmed.gla.ac.uk)

Version: 2 Date: 23 June 2010

Author's response to reviews:

thank you for asking me to resubmit this paper as a software article. I have done this, but am very uncertain as to whether this is appropriate, as the toolkit hardly counts as software. It uses an excel spread sheet. I am also concerned that the software paper subheadings do not include methods, which are an important part of this paper.

Best wishes

Elizabeth Murray